 Vc/17-May-05
Week 19  starting 9 May 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  09 May 05 @ 5/9/2005 1:40:30 PM   
	1
	A1
	MBAL1180
	PT 13.2 Resist. Meas. 
	297.41

	7
	A2
	MBBR3274
	PT 5.1 1.5 KA Quench LF
	2.01 HV

	11
	B1
	[image: image1.png]



	[image: image2.png]



	292.94

	10
	B2
	SSS122
	Prep 5 Cool Down
	203.11

	8
	C1
	MBBL1215
	Prep 5 Cool Down
	130.73

	2
	C2
	MBAL2073
	PT 6.5 Training 5
	4.55 HV

	4
	D1
	SSS009
	PT 1 IAP @ cold
	97.64

	3
	D2
	SSS158
	PT 6.1 Training 1
	1.91

	9
	E1
	MBBR3265
	ICS 4 Final connection
	294.01

	12
	E2
	[image: image3.png]
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	292.94

	6
	F1
	[image: image5.png]
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	292.94

	5
	F2
	MBBR3293
	Prep 5 Cool Down
	1.90


Temperature data are taken from TBSM18 site
Coordinator this week: stephane; MTM support on call :   16-3602 
15  complete Magnet Tests cycles   ( 2173 ROBTC)
09  dipoles( 1 with MM )+ 5  SSS( 0  with MM) +  1  SSSS (with MM)
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :   ?
· General Points 
· AOB

	Magnets under Test from 01-05-2005 to 07-05-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL2126 
	[image: image7.png]



	


	26-04-2005 
	35 h.
	90 h.
	19 h.
	18 h.
	19 h.
	04-05-2005 
	181 h.
	Stripping
	
	

	2
	TBA2
	MBAL2135 
	[image: image9.png]



	


	25-04-2005 
	38 h.
	31 h.
	68 h.
	19 h.
	7 h.
	02-05-2005 
	163 h.
	Stripping
	
	

	3
	TBA2
	MBAR3322 
	[image: image11.png]



	


	02-05-2005 
	28 h.
	23 h.
	18 h.
	15 h.
	11 h.
	05-05-2005 
	95 h.
	Stripping
	
	

	4
	TBB1
	SSS090 
	[image: image13.png]



	


	30-04-2005 
	25 h.
	15 h.
	16 h.
	8 h.
	18 h.
	04-05-2005 
	82 h.
	Stripping
	
	

	5
	TBB2
	SSS120 
	[image: image15.png]



	


	03-05-2005 
	31 h.
	12 h.
	23 h.
	10 h.
	17 h.
	07-05-2005 
	93 h.
	Stripping
	
	

	6
	TBB2
	SSS171 
	[image: image17.png]



	


	28-04-2005 
	38 h.
	14 h.
	39 h.
	10 h.
	10 h.
	03-05-2005 
	111 h.
	Stripping
	
	

	7
	TBC1
	MBAL3158 
	[image: image19.png]



	


	27-04-2005 
	35 h.
	32 h.
	19 h.
	13 h.
	7 h.
	01-05-2005 
	106 h.
	Stripping
	
	

	8
	TBC2
	MBAR3324 
	[image: image21.png]



	


	01-05-2005 
	21 h.
	41 h.
	21 h.
	16 h.
	9 h.
	05-05-2005 
	108 h.
	Stand By
	
	

	9
	TBD1
	SSS126 
	[image: image23.png]



	


	28-04-2005 
	31 h.
	14 h.
	19 h.
	7 h.
	17 h.
	02-05-2005 
	88 h.
	Stripping
	
	

	10
	TBD1
	SSS159 
	[image: image25.png]



	


	02-05-2005 
	39 h.
	15 h.
	17 h.
	14 h.
	8 h.
	06-05-2005 
	93 h.
	Stripping
	
	

	11
	TBD2
	SSS615 
	[image: image27.png]



	


	28-04-2005 
	44 h.
	65 h.
	35 h.
	8 h.
	53 h.
	06-05-2005 
	205 h.
	Stand By
	
	

	12
	TBE1
	MBAL2172 
	[image: image29.png]



	


	28-04-2005 
	27 h.
	23 h.
	28 h.
	16 h.
	7 h.
	03-05-2005 
	101 h.
	Stripping
	
	

	13
	TBE2
	MBAL2173 
	[image: image31.png]



	


	30-04-2005 
	24 h.
	43 h.
	40 h.
	15 h.
	0 h.
	05-05-2005 
	122 h.
	Not Found or Thermal cycle
	
	

	14
	TBF1
	MBAL1179 
	[image: image33.png]



	


	29-04-2005 
	31 h.
	24 h.
	36 h.
	14 h.
	9 h.
	04-05-2005 
	114 h.
	Stripping
	
	

	15
	TBF2
	MBBL2164 
	[image: image35.png]



	


	29-04-2005 
	38 h.
	52 h.
	38 h.
	15 h.
	13 h.
	06-05-2005 
	156 h.
	Stand By
	
	

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2126
	11723 A.
(D2-U)
	12491 A.
(D1- L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL2135
	11000 A.
(No Data)
	12064 A.
(No Data)
	12186 A.
(No Data)
	12444 A.
(No Data)
	12702 A.
(No Data)
	12850 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	MBAR3322
	12139 A.
(D1-L)
	12809 A.
(D2-L)
	12851 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS090
	12140 A.
(QI-34)
	12224 A.
(QE-34)
	12201 A.
(QE-34)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS120
	12039 A.
(QE-34)
	12484 A.
(QI-12)
	12732 A.
(QE_34)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS171
	12052 A.
(QE_34)
	12039.6 A.
(QE-34)
	12088.1 A.
(QE-34)
	12242 A.
(QI-34)
	12748 A.
(QI_12)
	12850 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-

	7
	MBAL3158
	12645 A.
(D2 - L)
	12807 A.
(D2-L)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR3324
	12231.5 A.
(D2-U)
	12715 A.
(D2-U)
	0 A.
(No Data)
	12850 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS126
	11850 A.
(QI_12)
	12498 A.
(QE-12)
	12672 A.
(QI-34)
	[image: image45.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS159
	12338.6 A.
(QI_34)
	12538.4 A.
(QE_12)
	12782 A.
(QI-12)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS615
	5489 A.
(Q_I_12)
	5820 A.
[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAL2172
	10314 A.
(D1-L)
	11922.6 A.
(D1_U)
	12509.8 A.
(D2-L)
	12795.4 A.
(D1_L)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL2173
	9408.8 A.
(D2-L)
	11068.8 A.
(D1-L)
	11941 A.
(D1-L)
	12050 A.
(D2-L)
	12259.4 A.
(D2-L)
	11785 A.
(D2_L)
	12423 A.
(D2_L)
	12177 A.
(D2-L)
	12699 A.
(D2-L)
	[image: image49.png]



	-
	-
	-

	14
	MBAL1179
	9858 A.
(D1-L)
	11507 A.
(D2-L)
	11941 A.
(D1-L)
	12020 A.
(D1-U)
	12271 A.
(D2-L)
	12575 A.
(D2-L)
	12819 A.
(D1-U)
	12860 A.
[image: image50.jpg]



	-
	-
	-
	-
	-

	15
	MBBL2164
	12553 A.
(D2-U-S02)
	12900 A.
[image: image51.jpg]



	12850 A.
[image: image52.jpg]



	12850 A.
[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Week 17  :  01May to 07 may 05  (15 Magnet Tests)
[    On bench rapid thermal cycle   ]
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday _04_A

	TBD2

	Stand by of SIMCO survey team reserved till Wednesday 17h00.
	MTM
	
	TC called  Guy who accepted to make the survey in the Aperture 2 of the SSS615

	Wednesday _04_N
	TBA1
	Gamelle missing. ICS cannot close the magnet
	MTM
	
	

	Thursday _05_A
	
	HV mobile rack 92 has a bad D21 connector.
	MTM
	
	Monday Richard replaced the wrong module mounted in connector D20. Rack is back in operation.

	Thursday _05_N
	TBB1
	Temperatures not correct on Video display for all the benches.
	ACR
	
	Jerome Axensalva found that a server was down Problem solved..

	Friday _06_M
	TBA1
	Bad alignment of the shafts with the TRU.
	MTM
	
	Didier found that the table of the TRU had been moved laterally. He corrected the position of the table. ICS should act on the yellow trolley.

	Friday _06_M
	TBE1
	HV at cold : temperature sensor vs ground failed on MB3107.
	MTM
	
	Salem found that a pin was bent in the temp. sensor connector in the IFS box. He repaired the connector and the HV test passed OK

	Friday _06_A
	TBB
	Tema is not working :No communication possible,LD_LIBRARY_PATH environment variable missing
	AB/CO
	TEMA
	Salem contacted Adrian. PC next to the sun was moved and the network cable was unplugged

	Friday _06_N
	TBD2
	SSS158: No response for lf DAQ.
	ICS?
	
	Salem asked to trigger the DAQ using the card in the safety matrix: no trigger.

The bench was in the special SSS configuration.

	Saturday_07_M
	TBD1
	While launching Prep 2.2 (IAP@warm), PLCSE indicated that the QH are not charged, But actually heaters were charged (approx. 840V).
	MTM
	
	Salem found a  bad communication between PLCSE and SERVER application. He solved the problem by resetting server and PLC.  

	Saturday_07_M
	TBA1
	Over-range in AP2 during MM2
	MTM
	
	Salem found the origin of the problem: end cover touching the flat cable.

	Saturday_07_M
	TBA1
	Waterflow problem. PLC for water-flow control not responding, Quench could not be launched.
	MTM
	
	Salem resetted the PLCCU: OK

	Saturday_07_N
	TBA1
	Cryo loss during LHC cycle due to a rapid drop in helium flow to the B current lead.

SPA during the pre-cycle.
	MTM
	
	LHC cycle to be repeated.

	Saturday_07_N
	TBF2
	Cryo team received a high temperature warning (65 deg C) on the anticryostat L1

when the cables were connected. 

Alarm on the anticryostat.
	MTM

MTM
	
	Wrong parameter in the L1 anticryostat?

Salem updated the parameters from MTF.



	Sunday_08_M
	TBC1
	Bad manipulation during C1 connection, General cut of electricity

Ext trigger interlock on RMA


	ICS
	
	A1 was running with a LHC cycle, the cycle was stopped., RMA also switched off.

TCR will send someone for repair. Salem solved it by switching on extraction switch

	Sunday_08_M
	TBA1
	Valve opened but no water.


	MTM
	
	Salem and TCR called.

Salem solved the problem by resetting PLC-CU and



	Sunday_08_A

Mondat_09_M
	TBD1
	IAP @cold on SSS09: spurious output trigger during AC compensation.


	MTM
	
	Investigation by Salem, Richard,Giorgio. Problem coming from the magnet, VTaps inverted inside. It should have been detected with IAP @warm.

	Sunday_08_M
	TBC1
	Login problem on cluster C due to an incorrect desktop (application not running)


	AB/CO
	
	M.Marchesotti solved the problem











